Author's response to reviews

Title: Association between anti-Porphyromonas gingivalis or anti-alpha-enolase antibody and severity of periodontitis or rheumatoid arthritis (RA) disease activity in RA

Authors:

Joo Youn Lee (janusx@snu.ac.kr)
In Ah Choi (jachoi@outlook.com)
Jin-Hee Kim (2001aroma@hanmail.net)
Kyoung-Hwa Kim (khk6911@hanmail.net)
Eun Young Lee (elee@snu.ac.kr)
Eun Bong Lee (led7616@snu.ac.kr)
Yong-Moo Lee (ymlee@snu.ac.kr)
Yeong Wook Song (ysong@snu.ac.kr)

Version: 2 Date: 13 February 2015

Author's response to reviews: see over
Editor

BMC Musculoskeletal Disorders

Dear Editor:

Enclosed is a manuscript "Association between anti-Porphyromonas gingivalis or anti-α-enolase antibody and severity of periodontitis or rheumatoid arthritis (RA) disease activity in RA" by JY Lee et al. We measured serum antibody against P. gingivalis and human ENO1 in patients with RA and investigated their association with the severity of PD or disease activity of RA. Anti-P. gingivalis and anti-ENO1 antibody titers were correlated with the severity of PD in RA. Anti-ENO1 antibody titers, but not anti-P. gingivalis antibody titers, were further associated with RA disease activity.

The paper is submitted to be considered for publication as an original article in your journal. The manuscript has been seen and approved by all the authors and they have given necessary attention to ensure the integrity of the work. Neither the entire paper nor any part of its content has been published or has been accepted by another journal. The paper is not being submitted to any other journal. Thank you for your attention to our paper.

Yours sincerely,

Yeong Wook Song, MD
Department of Internal Medicine, Seoul National University Hospital
101 Daehak-ro, Jongno-gu, Seoul 110-744, South Korea
Tel: +82-2-2072-2234
Fax: +82-2-762-9662
E-mail: ysong@snu.ac.kr